\begin{table}%t1
%%\centering

\par
\caption{\label{literature}The [Fe/H] values from photometric calibrations for open clusters found in the literature.}
\begin{tabular}{cccccc|cccccc}
\hline
\hline
Cluster	&	[Fe/H]	&	$\sigma$[Fe/H] &	Stars	&	System	&	Ref.	&	Cluster	&	[Fe/H]	&	$\sigma$[Fe/H]	&	Stars	&	
System	&	Ref.	\\
\hline
Berkeley 1	&	$-$0.14	&	--	&	--	&	J	&	[57] &	IC 4651	&	+0.18	&	0.05	&	10	&	S	&	[41] \\
Berkeley 2	&	$-$0.59	&	--	&	--	&	J	&	[57] &	IC 4651	&	$-$0.09	&	0.03	&	--	&	J	&	[43] \\
Berkeley 7	&	$-$0.25	&	--	&	--	&	J	&	[57] &	IC 4651	&	+0.08	&	0.09	&	8	&	D	&	[49] \\
Berkeley 14	&	$-$1.20	&	--	&	--	&	J	&	[57] &	IC 4651	&	$-$0.14	&	--	&	--	&	J	&	[57] \\
Berkeley 21	&	$-$0.96	&	--	&	--	&	J	&	[57] &	IC 4651	&	+0.095	&	0.053	&	11	&	D	&	[61] \\
Berkeley 22	&	$-$0.42	&	--	&	7	&	J	&	[43] &	IC 4665	&	$-$0.11	&	0.07	&	5	&	S	&	[40] \\
Berkeley 28	&	$-$0.61	&	--	&	--	&	J	&	[57] &	IC 4725	&	$-$0.092	&	--	&	63	&	J	&	[7] \\
Berkeley 29	&	$-$0.30	&	--	&	19	&	J	&	[43] &	IC 4756	&	+0.003	&	--	&	36	&	J	&	[7] \\
Berkeley 30	&	+0.10	&	--	&	--	&	J	&	[57] &	IC 4756	&	+0.04	&	0.05	&	10	&	D	&	[54] \\
Berkeley 31	&	$-$0.60	&	--	&	--	&	J	&	[57] &	IC 4756	&	$-$0.012	&	0.072	&	6	&	D	&	[61] \\
Berkeley 32	&	$-$0.37	&	0.05	&	--	&	J, W	&	[32] &	IC 4996	&	$-$0.47	&	--	&	--	&	J	&	[57] \\
Berkeley 32	&	$-$0.37	&	0.04	&	--	&	J	&	[43] &	King 2	&	$-$0.32	&	--	&	--	&	J	&	[57] \\
Berkeley 32	&	$-$0.55	&	--	&	--	&	J	&	[57] &	King 7	&	+0.03	&	--	&	--	&	J	&	[57] \\
Berkeley 39	&	$-$0.42	&	0.11	&	8	&	W	&	[29] &	King 10	&	+0.09	&	--	&	--	&	J	&	[57] \\
Berkeley 39	&	$-$0.26	&	--	&	27	&	J	&	[43] &	King 11	&	$-$0.31	&	--	&	21	&	J	&	[43] \\
Berkeley 39	&	$-$0.33	&	--	&	--	&	J	&	[57] &	King 11	&	$-$0.37	&	--	&	--	&	J	&	[57] \\
Berkeley 42	&	+0.08	&	--	&	--	&	J	&	[57] &	Loden 807	&	$-$0.19	&	0.06	&	2	&	D	&	[25] \\
Berkeley 58	&	+0.09	&	--	&	--	&	J	&	[57] &	Loden 807	&	$-$0.22	&	--	&	1	&	D	&	[49] \\
Berkeley 60	&	$-$0.77	&	--	&	--	&	J	&	[57] &	Loden 807	&	$-$0.130	&	0.090	&	1	&	D	&	[61] \\
Berkeley 62	&	$-$0.85	&	--	&	--	&	J	&	[57] &	Melotte 20	&	+0.07	&	--	&	119	&	J	&	[7] \\
Berkeley 79	&	$-$0.32	&	--	&	--	&	J	&	[57] &	Melotte 20	&	+0.04	&	0.02	&	19	&	S	&	[40] \\
Berkeley 86	&	$-$0.08	&	--	&	--	&	J	&	[57] &	Melotte 20	&	+0.01	&	0.09	&	19	&	S	&	[41] \\
Collinder 74	&	+0.05	&	--	&	--	&	J	&	[57] &	Melotte 22	&	+0.08	&	--	&	31	&	J	&	[7] \\
Collinder 140	&	$-$0.092	&	--	&	21	&	J	&	[7] &	Melotte 22	&	+0.05	&	0.02	&	30	&	S	&	[40] \\
Collinder 140	&	+0.00	&	0.20	&	--	&	S	&	[37] &	Melotte 22	&	+0.06	&	0.12	&	23	&	S	&	[41] \\
Collinder 173	&	$-$0.06	&	0.20	&	--	&	S	&	[37] &	Melotte 25	&	+0.08	&	--	&	48	&	J	&	[7] \\
Collinder 258	&	+0.07	&	0.21	&	2	&	D	&	[17] &	Melotte 25	&	+0.13	&	0.01	&	49	&	S	&	[40] \\
Collinder 258	&	$-$0.24	&	--	&	1	&	D	&	[49] &	Melotte 25	&	+0.12	&	0.09	&	42	&	S	&	[41] \\
Collinder 258	&	$-$0.090	&	0.090	&	1	&	D	&	[61] &	Melotte 25	&	+0.19	&	0.03	&	4	&	D	&	[49] \\
Collinder 272	&	$-$0.40	&	--	&	--	&	J	&	[57] &	Melotte 25	&	+0.175	&	0.034	&	4	&	D	&	[61] \\
Haffner 8	&	$-$0.09	&	0.10	&	2	&	D, W	&	[17] &	Melotte 66	&	$-$0.35	&	0.20	&	4	&	W	&	[28] \\
Haffner 8	&	+0.06	&	0.06	&	2	&	D	&	[49] &	Melotte 66	&	$-$0.52	&	0.05	&	--	&	J	&	[43] \\
Haffner 8	&	+0.060	&	0.040	&	2	&	D	&	[61] &	Melotte 66	&	$-$0.44	&	0.09	&	6	&	D	&	[49] \\
IC 348	&	+0.03	&	--	&	--	&	J	&	[57] &	Melotte 66	&	$-$0.53	&	0.08	&	--	&	S	&	[60] \\
IC 1311	&	$-$0.23	&	--	&	--	&	J	&	[57] &	Melotte 66	&	$-$0.349	&	0.145	&	7	&	D	&	[61] \\
IC 1590	&	$-$0.73	&	--	&	--	&	J	&	[57] &	Melotte 71	&	$-$0.57	&	0.18	&	14	&	W	&	[29] \\
IC 1805	&	+0.08	&	--	&	--	&	J	&	[57] &	Melotte 71	&	$-$0.17	&	0.02	&	--	&	C, S	&	[62] \\
IC 2391	&	$-$0.15	&	--	&	--	&	S	&	[27] &	Melotte 105	&	+0.00	&	--	&	--	&	J	&	[57] \\
IC 2391	&	$-$0.09	&	0.20	&	--	&	S	&	[37] &	Melotte 111	&	$-$0.063	&	--	&	30	&	J	&	[7] \\
IC 2488	&	+0.095	&	0.08	&	3	&	D, W	&	[22] &	Melotte 111	&	$-$0.03	&	0.02	&	18	&	S	&	[40] \\
IC 2602	&	$-$0.232	&	--	&	--	&	J	&	[7] &	Melotte 111	&	$-$0.01	&	0.07	&	17	&	S	&	[41] \\
IC 2714	&	$-$0.12	&	0.09	&	19	&	D, W	&	[19] &	NGC 103	&	$-$0.85	&	--	&	--	&	J	&	[57] \\
IC 2714	&	+0.01	&	0.10	&	10	&	D	&	[49] &	NGC 129	&	$-$0.46	&	--	&	--	&	J	&	[57] \\
IC 2714	&	$-$0.011	&	0.116	&	5	&	D	&	[61] &	NGC 188	&	+0.055	&	--	&	34	&	J	&	[7] \\
IC 4651	&	+0.077	&	0.012	&	9	&	S	&	[1] &	NGC 188	&	$-$0.16	&	0.11	&	7	&	D	&	[49] \\
IC 4651	&	$-$0.194	&	--	&	96	&	J	&	[7] &	NGC 188	&	$-$0.046	&	0.098	&	8	&	D	&	[61] \\
IC 4651	&	+0.21	&	0.03	&	31	&	D	&	[10] &	NGC 366	&	$-$0.55	&	--	&	--	&	J	&	[57] \\
IC 4651	&	+0.15	&	0.15	&	--	&	W	&	[10] &	NGC 381	&	$-$0.04	&	--	&	--	&	J	&	[57] \\
IC 4651	&	+0.11	&	0.05	&	--	&	D, W	&	[17] &	NGC 433	&	$-$0.68	&	--	&	--	&	J	&	[57] \\
\par
NGC 436	&	$-$0.77	&	--	&	--	&	J	&	[57] &	NGC 2301	&	+0.060	&	0.060	&	2	&	D	&	[61] \\
NGC 457	&	$-$0.46	&	--	&	--	&	J	&	[57] &	NGC 2324	&	$-$0.163	&	--	&	64	&	J	&	[7] \\
NGC 581	&	$-$0.85	&	--	&	--	&	J	&	[57] &	NGC 2324	&	$-$1.01	&	0.27	&	9	&	W	&	[29] \\
NGC 654	&	$-$0.68	&	--	&	--	&	J	&	[57] &	NGC 2324	&	$-$0.31	&	0.04	&	5	&	W	&	[52] \\
NGC 663	&	$-$0.70	&	--	&	--	&	J	&	[57] &	NGC 2335	&	+0.22	&	--	&	--	&	D	&	[9] \\
NGC 752	&	$-$0.06	&	0.03	&	10	&	C	&	[3] &	NGC 2335	&	+0.02	&	--	&	1	&	D	&	[49] \\
NGC 752	&	$-$0.311	&	--	&	14	&	J	&	[7] &	NGC 2335	&	$-$0.030	&	0.090	&	1	&	D	&	[61] \\
NGC 752	&	$-$0.26	&	0.06	&	7	&	D	&	[20] &	NGC 2343	&	$-$0.15	&	--	&	--	&	D	&	[9] \\
NGC 752	&	$-$0.02	&	0.02	&	29	&	S	&	[40] &	NGC 2343	&	$-$0.39	&	--	&	1	&	D	&	[49] \\
NGC 752	&	$-$0.05	&	0.13	&	26	&	S	&	[41] &	NGC 2354	&	$-$0.30	&	0.02	&	12	&	D, J, W	&	[21] \\
NGC 752	&	$-$0.22	&	0.05	&	7	&	D	&	[49] &	NGC 2355	&	+0.13	&	--	&	50	&	J	&	[34] \\
NGC 752	&	$-$0.21	&	0.09	&	19	&	D, J, S	&	[58] &	NGC 2355	&	$-$0.14	&	--	&	--	&	J	&	[57] \\
NGC 752	&	$-$0.160	&	0.040	&	6	&	D	&	[61] &	NGC 2360	&	$-$0.092	&	--	&	34	&	J	&	[7] \\
NGC 1039	&	$-$0.291	&	--	&	36	&	J	&	[7] &	NGC 2360	&	$-$0.12	&	0.03	&	13	&	D	&	[25] \\
NGC 1193	&	$-$0.45	&	0.12	&	32	&	J	&	[36] &	NGC 2360	&	$-$0.29	&	0.04	&	6	&	W	&	[29] \\
NGC 1193	&	$-$0.54	&	--	&	16	&	J	&	[43] &	NGC 2360	&	$-$0.14	&	0.07	&	10	&	D	&	[49] \\
NGC 1342	&	$-$0.163	&	--	&	14	&	J	&	[7] &	NGC 2360	&	$-$0.154	&	0.113	&	11	&	D	&	[61] \\
NGC 1342	&	$-$0.24	&	0.12	&	3	&	D	&	[49] &	NGC 2420	&	$-$0.37	&	0.05	&	106	&	C, S	&	[5] \\
NGC 1528	&	$-$0.133	&	--	&	17	&	J	&	[7] &	NGC 2420	&	$-$0.43	&	0.07	&	--	&	J	&	[43] \\
NGC 1545	&	$-$0.16	&	--	&	1	&	D	&	[20] &	NGC 2420	&	$-$0.43	&	0.10	&	8	&	D	&	[49] \\
NGC 1545	&	$-$0.18	&	--	&	1	&	D	&	[49] &	NGC 2420	&	$-$0.281	&	0.073	&	10	&	D	&	[61] \\
NGC 1545	&	$-$0.060	&	0.090	&	1	&	D	&	[61] &	NGC 2422	&	+0.11	&	0.10	&	11	&	S	&	[41] \\
NGC 1662	&	$-$0.232	&	--	&	20	&	J	&	[7] &	NGC 2423	&	+0.00	&	0.08	&	4	&	D, W	&	[17] \\
NGC 1662	&	$-$0.13	&	0.02	&	2	&	D	&	[20] &	NGC 2423	&	+0.03	&	0.03	&	8	&	D	&	[25] \\
NGC 1662	&	$-$0.03	&	0.01	&	2	&	D	&	[49] &	NGC 2423	&	+0.11	&	0.04	&	7	&	D	&	[49] \\
NGC 1662	&	$-$0.095	&	0.015	&	2	&	D	&	[61] &	NGC 2423	&	+0.143	&	0.092	&	3	&	D	&	[61] \\
NGC 1750	&	$-$0.69	&	--	&	--	&	J	&	[57] &	NGC 2437	&	$-$0.16	&	0.11	&	2	&	D	&	[25] \\
NGC 1798	&	$-$0.46	&	--	&	--	&	J	&	[57] &	NGC 2437	&	+0.07	&	--	&	1	&	D	&	[49] \\
NGC 1817	&	$-$0.291	&	--	&	17	&	J	&	[7] &	NGC 2447	&	$-$0.10	&	0.08	&	9	&	D, J, W	&	[23] \\
NGC 1817	&	$-$0.33	&	0.08	&	15	&	W	&	[45] &	NGC 2451	&	$-$0.50	&	0.02	&	--	&	S	&	[37] \\
NGC 1931	&	$-$0.06	&	--	&	--	&	J	&	[57] &	NGC 2451	&	$-$0.106	&	0.122	&	--	&	D	&	[46] \\
NGC 2099	&	+0.18	&	0.07	&	3	&	D	&	[49] &	NGC 2477	&	$-$0.008	&	--	&	29	&	J	&	[7] \\
NGC 2099	&	+0.089	&	0.146	&	5	&	D	&	[61] &	NGC 2477	&	$-$0.13	&	0.18	&	16	&	W	&	[29] \\
NGC 2112	&	$-$1.30	&	0.22	&	4	&	S	&	[53] &	NGC 2477	&	+0.04	&	0.01	&	--	&	D	&	[55] \\
NGC 2158	&	$-$0.36	&	0.18	&	3	&	D	&	[49] &	NGC 2477	&	$-$0.05	&	--	&	--	&	J	&	[57] \\
NGC 2158	&	$-$0.238	&	0.064	&	5	&	D	&	[61] &	NGC 2482	&	+0.20	&	0.01	&	--	&	D	&	[10] \\
NGC 2168	&	$-$0.28	&	--	&	1	&	D	&	[49] &	NGC 2482	&	+0.10	&	0.10	&	--	&	W	&	[10] \\
NGC 2168	&	$-$0.39	&	--	&	--	&	J	&	[57] &	NGC 2482	&	+0.07	&	0.07	&	3	&	D, W	&	[17] \\
NGC 2168	&	$-$0.160	&	0.090	&	1	&	D	&	[61] &	NGC 2482	&	+0.14	&	0.04	&	3	&	D	&	[25] \\
NGC 2192	&	$-$0.22	&	--	&	--	&	J	&	[57] &	NGC 2482	&	+0.13	&	0.04	&	3	&	D	&	[49] \\
NGC 2194	&	$-$0.27	&	0.06	&	9	&	W	&	[50] &	NGC 2482	&	+0.120	&	0.026	&	3	&	D	&	[61] \\
NGC 2204	&	$-$0.618	&	--	&	9	&	J	&	[7] &	NGC 2489	&	+0.00	&	--	&	1	&	D	&	[20] \\
NGC 2204	&	$-$0.41	&	0.19	&	6	&	D	&	[26] &	NGC 2489	&	+0.10	&	--	&	1	&	D	&	[49] \\
NGC 2204	&	$-$0.39	&	0.22	&	5	&	D	&	[49] &	NGC 2489	&	+0.01	&	--	&	--	&	J	&	[57] \\
NGC 2204	&	$-$0.338	&	0.250	&	5	&	D	&	[61] &	NGC 2489	&	+0.080	&	0.010	&	2	&	D	&	[61] \\
NGC 2232	&	$-$0.205	&	0.213	&	--	&	D	&	[46] &	NGC 2506	&	$-$0.58	&	0.14	&	23	&	W	&	[29] \\
NGC 2236	&	$-$0.30	&	0.20	&	13	&	W	&	[24] &	NGC 2506	&	$-$0.57	&	--	&	16	&	D	&	[38] \\
NGC 2243	&	$-$0.57	&	0.03	&	100	&	C, S	&	[4] &	NGC 2506	&	$-$0.48	&	0.08	&	4	&	D	&	[49] \\
NGC 2243	&	$-$0.65	&	0.12	&	4	&	D	&	[49] &	NGC 2506	&	$-$0.58	&	--	&	--	&	J	&	[57] \\
NGC 2243	&	$-$0.480	&	0.160	&	5	&	D	&	[61] &	NGC 2506	&	$-$0.368	&	0.108	&	5	&	D	&	[61] \\
NGC 2244	&	$-$0.46	&	--	&	--	&	J	&	[57] &	NGC 2516	&	$-$0.422	&	--	&	67	&	J	&	[7] \\
NGC 2251	&	$-$0.25	&	0.04	&	3	&	W	&	[45] &	NGC 2516	&	+0.00	&	0.10	&	2	&	D	&	[17] \\
NGC 2251	&	$-$0.17	&	0.06	&	3	&	D	&	[49] &	NGC 2516	&	$-$0.28	&	0.20	&	--	&	S	&	[37] \\
NGC 2264	&	+0.00	&	--	&	--	&	D	&	[9] &	NGC 2516	&	+0.06	&	0.06	&	8	&	S	&	[41] \\
NGC 2264	&	$-$0.09	&	0.30	&	--	&	S	&	[48] &	NGC 2516	&	+0.02	&	--	&	1	&	D	&	[49] \\
NGC 2264	&	$-$0.16	&	--	&	--	&	J	&	[57] &	NGC 2516	&	$-$0.10	&	0.04	&	--	&	J	&	[56] \\
NGC 2266	&	$-$0.26	&	0.20	&	9	&	W	&	[33] &	NGC 2516	&	+0.060	&	0.030	&	2	&	D	&	[61] \\
NGC 2281	&	$-$0.074	&	--	&	38	&	J	&	[7] &	NGC 2527	&	$-$0.02	&	--	&	--	&	D	&	[9] \\
NGC 2287	&	+0.065	&	--	&	14	&	J	&	[7] &	NGC 2527	&	$-$0.09	&	--	&	1	&	D	&	[49] \\
NGC 2287	&	$-$0.10	&	0.11	&	10	&	S	&	[41] &	NGC 2527	&	$-$0.080	&	0.090	&	1	&	D	&	[61] \\
NGC 2287	&	$-$0.246	&	0.255	&	--	&	D	&	[46] &	NGC 2539	&	+0.24	&	0.06	&	--	&	D	&	[12] \\
NGC 2287	&	$-$0.13	&	0.06	&	3	&	D	&	[49] &	NGC 2539	&	+0.00	&	0.20	&	--	&	W	&	[12] \\
NGC 2287	&	+0.040	&	0.022	&	4	&	D	&	[61] &	NGC 2539	&	+0.03	&	0.09	&	7	&	D, W	&	[17] \\
NGC 2301	&	+0.014	&	--	&	18	&	J	&	[7] &	NGC 2539	&	+0.17	&	0.08	&	3	&	D	&	[49] \\
NGC 2301	&	+0.10	&	0.05	&	2	&	D	&	[17] &	NGC 2539	&	+0.137	&	0.062	&	6	&	D	&	[61] \\
NGC 2301	&	+0.04	&	0.12	&	13	&	S	&	[41] &	NGC 2546	&	+0.26	&	--	&	--	&	D	&	[9] \\
NGC 2301	&	+0.01	&	--	&	1	&	D	&	[49] &	NGC 2546	&	+0.11	&	+0.09	&	2	&	D	&	[25] \\
NGC 2546	&	$-$0.14	&	--	&	1	&	D	&	[49] &	NGC 4755	&	$-$0.21	&	--	&	--	&	J	&	[57] \\
NGC 2546	&	+0.120	&	0.130	&	2	&	D	&	[61] &	NGC 4815	&	$-$0.19	&	--	&	--	&	J	&	[57] \\
NGC 2547	&	$-$0.121	&	--	&	43	&	J	&	[7] &	NGC 5138	&	+0.18	&	--	&	1	&	D	&	[49] \\
NGC 2547	&	$-$0.13	&	--	&	1	&	D	&	[8] &	NGC 5138	&	+0.120	&	0.050	&	2	&	D	&	[61] \\
NGC 2547	&	$-$0.18	&	0.20	&	--	&	S	&	[37] &	NGC 5168	&	+0.05	&	--	&	--	&	J	&	[57] \\
NGC 2547	&	$-$0.21	&	--	&	1	&	D	&	[49] &	NGC 5281	&	$-$0.18	&	--	&	1	&	D	&	[17] \\
NGC 2547	&	$-$0.160	&	0.090	&	1	&	D	&	[61] &	NGC 5316	&	+0.19	&	0.11	&	--	&	D, J, W	&	[14] \\
NGC 2548	&	+0.14	&	--	&	--	&	D	&	[9] &	NGC 5316	&	$-$0.02	&	0.12	&	2	&	D	&	[49] \\
NGC 2548	&	+0.01	&	0.02	&	2	&	D	&	[49] &	NGC 5316	&	+0.128	&	0.127	&	4	&	D	&	[61] \\
NGC 2548	&	+0.080	&	0.020	&	3	&	D	&	[61] &	NGC 5606	&	+0.095	&	--	&	--	&	J	&	[57] \\
NGC 2567	&	+0.00	&	--	&	--	&	D	&	[13] &	NGC 5617	&	+0.31	&	0.10	&	2	&	D, W	&	[17] \\
NGC 2567	&	+0.00	&	0.10	&	--	&	W	&	[13] &	NGC 5617	&	$-$0.32	&	--	&	--	&	J	&	[57] \\
NGC 2567	&	$-$0.09	&	0.19	&	2	&	D	&	[49] &	NGC 5662	&	$-$0.03	&	0.13	&	2	&	D, W	&	[18] \\
NGC 2567	&	+0.22	&	--	&	--	&	J	&	[57] &	NGC 5662	&	+0.09	&	--	&	--	&	J	&	[57] \\
NGC 2567	&	$-$0.030	&	0.120	&	2	&	D	&	[61] &	NGC 5822	&	+0.01	&	0.04	&	--	&	D	&	[11] \\
NGC 2571	&	+0.05	&	--	&	23	&	J	&	[7] &	NGC 5822	&	$-$0.25	&	0.15	&	--	&	W	&	[11] \\
NGC 2627	&	$-$0.12	&	0.08	&	8	&	W	&	[51] &	NGC 5822	&	$-$0.12	&	0.08	&	14	&	D, W	&	[17] \\
NGC 2627	&	$-$0.04	&	--	&	--	&	J	&	[57] &	NGC 5822	&	+0.03	&	0.09	&	13	&	D	&	[49] \\
NGC 2632	&	+0.04	&	--	&	76	&	J	&	[7] &	NGC 5822	&	$-$0.15	&	0.015	&	--	&	D, J	&	[59] \\
NGC 2632	&	+0.20	&	0.07	&	3	&	D	&	[20] &	NGC 5822	&	$-$0.020	&	0.086	&	14	&	D	&	[61] \\
NGC 2632	&	+0.09	&	0.02	&	47	&	S	&	[40] &	NGC 6025	&	+0.195	&	--	&	48	&	J	&	[7] \\
NGC 2632	&	+0.10	&	0.15	&	42	&	S	&	[41] &	NGC 6067	&	+0.10	&	0.06	&	6	&	D, W	&	[17] \\
NGC 2632	&	+0.19	&	0.06	&	3	&	D	&	[49] &	NGC 6067	&	$-$0.01	&	0.07	&	3	&	D	&	[49] \\
NGC 2632	&	+0.142	&	0.069	&	4	&	D	&	[61] &	NGC 6067	&	+0.138	&	0.064	&	5	&	D	&	[61] \\
NGC 2658	&	$-$0.94	&	--	&	--	&	J	&	[57] &	NGC 6087	&	$-$0.04	&	--	&	--	&	J	&	[57] \\
NGC 2660	&	$-$1.05	&	0.16	&	7	&	W	&	[29] &	NGC 6134	&	+0.28	&	0.02	&	--	&	S	&	[6] \\
NGC 2660	&	$-$0.27	&	0.13	&	5	&	D	&	[49] &	NGC 6134	&	$-$0.05	&	0.12	&	17	&	D, J, W	&	[15] \\
NGC 2660	&	$-$0.05	&	--	&	--	&	J	&	[57] &	NGC 6134	&	+0.20	&	0.10	&	7	&	D	&	[49] \\
NGC 2660	&	$-$0.181	&	0.118	&	5	&	D	&	[61] &	NGC 6134	&	+0.30	&	--	&	--	&	J	&	[57] \\
NGC 2669	&	$-$0.20	&	--	&	1	&	D	&	[17] &	NGC 6134	&	+0.182	&	0.087	&	10	&	D	&	[61] \\
NGC 2682	&	$-$0.029	&	--	&	58	&	J	&	[7] &	NGC 6192	&	$-$0.10	&	0.09	&	--	&	S	&	[47] \\
NGC 2682	&	$-$0.05	&	0.03	&	22	&	D	&	[31] &	NGC 6192	&	$-$0.35	&	--	&	--	&	J	&	[57] \\
NGC 2682	&	$-$0.06	&	0.07	&	36	&	S	&	[42] &	NGC 6204	&	$-$0.14	&	--	&	--	&	J	&	[57] \\
NGC 2682	&	$-$0.05	&	0.04	&	--	&	J	&	[43] &	NGC 6208	&	$-$0.03	&	0.06	&	3	&	D	&	[49] \\
NGC 2682	&	$-$0.01	&	0.11	&	14	&	D	&	[49] &	NGC 6231	&	+0.26	&	--	&	--	&	J	&	[57] \\
NGC 2682	&	$-$0.11	&	--	&	--	&	J	&	[57] &	NGC 6259	&	+0.06	&	0.08	&	8	&	W	&	[39] \\
NGC 2682	&	+0.000	&	0.092	&	25	&	D	&	[61] &	NGC 6259	&	+0.01	&	0.09	&	2	&	D	&	[49] \\
NGC 2818	&	$-$0.31	&	0.13	&	11	&	W	&	[29] &	NGC 6281	&	+0.07	&	--	&	37	&	J	&	[7] \\
NGC 2818	&	$-$0.29	&	--	&	--	&	J	&	[57] &	NGC 6281	&	$-$0.08	&	0.08	&	2	&	D, W	&	[17] \\
NGC 2910	&	$-$0.04	&	--	&	--	&	J	&	[57] &	NGC 6281	&	+0.00	&	0.12	&	2	&	D	&	[49] \\
NGC 2972	&	$-$0.15	&	0.08	&	2	&	D	&	[17] &	NGC 6281	&	+0.005	&	0.095	&	2	&	D	&	[61] \\
NGC 2972	&	$-$0.09	&	0.01	&	2	&	D	&	[49] &	NGC 6405	&	+0.07	&	--	&	20	&	J	&	[7] \\
NGC 2972	&	$-$0.073	&	0.021	&	3	&	D	&	[61] &	NGC 6425	&	+0.25	&	0.04	&	2	&	D	&	[25] \\
NGC 3114	&	$-$0.04	&	0.04	&	7	&	D, W	&	[17] &	NGC 6425	&	+0.09	&	--	&	1	&	D	&	[49] \\
NGC 3114	&	$-$0.14	&	0.14	&	4	&	D	&	[49] &	NGC 6451	&	$-$0.34	&	0.08	&	--	&	S	&	[47] \\
NGC 3114	&	+0.022	&	0.070	&	5	&	D	&	[61] &	NGC 6451	&	$-$0.20	&	--	&	--	&	J	&	[57] \\
NGC 3532	&	+0.08	&	0.08	&	5	&	W	&	[16] &	NGC 6475	&	+0.02	&	0.07	&	11	&	S	&	[41] \\
NGC 3532	&	$-$0.10	&	0.09	&	3	&	D	&	[49] &	NGC 6475	&	+0.03	&	--	&	1	&	D	&	[49] \\
NGC 3532	&	$-$0.022	&	0.088	&	5	&	D	&	[61] &	NGC 6475	&	+0.070	&	0.090	&	1	&	D	&	[61] \\
NGC 3680	&	$-$0.14	&	0.03	&	30	&	C, S	&	[2] &	NGC 6494	&	$-$0.14	&	0.15	&	--	&	D, J, W	&	[14] \\
NGC 3680	&	+0.09	&	0.02	&	--	&	S	&	[6] &	NGC 6494	&	+0.13	&	0.15	&	2	&	D	&	[49] \\
NGC 3680	&	$-$0.145	&	--	&	27	&	J	&	[7] &	NGC 6494	&	+0.090	&	0.117	&	4	&	D	&	[61] \\
NGC 3680	&	+0.00	&	0.04	&	--	&	D	&	[10] &	NGC 6520	&	$-$0.25	&	--	&	--	&	J	&	[57] \\
NGC 3680	&	+0.10	&	0.20	&	--	&	W	&	[10] &	NGC 6611	&	$-$0.46	&	--	&	--	&	J	&	[57] \\
NGC 3680	&	+0.02	&	0.03	&	8	&	D, W	&	[17] &	NGC 6633	&	$-$0.133	&	--	&	39	&	J	&	[7] \\
NGC 3680	&	+0.09	&	0.08	&	25	&	S	&	[41] &	NGC 6633	&	$-$0.01	&	0.03	&	5	&	D	&	[25] \\
NGC 3680	&	$-$0.16	&	0.07	&	8	&	D	&	[49] &	NGC 6633	&	$-$0.02	&	0.05	&	3	&	D	&	[49] \\
NGC 3680	&	$-$0.121	&	0.051	&	8	&	D	&	[61] &	NGC 6649	&	+0.05	&	--	&	--	&	J	&	[57] \\
NGC 3766	&	$-$0.47	&	--	&	--	&	J	&	[57] &	NGC 6705	&	+0.07	&	--	&	52	&	J	&	[7] \\
NGC 3960	&	$-$0.68	&	0.28	&	11	&	W	&	[29] &	NGC 6705	&	$-$0.39	&	--	&	--	&	J	&	[57] \\
NGC 3960	&	$-$0.06	&	0.05	&	3	&	D	&	[49] &	NGC 6716	&	$-$0.311	&	--	&	24	&	J	&	[7] \\
NGC 3960	&	$-$0.170	&	0.131	&	6	&	D	&	[61] &	NGC 6716	&	$-$0.26	&	--	&	--	&	J	&	[57] \\
NGC 4103	&	$-$0.47	&	--	&	--	&	J	&	[57] &	NGC 6755	&	$-$0.03	&	--	&	--	&	J	&	[57] \\
NGC 4349	&	$-$0.23	&	0.08	&	--	&	D, J, W	&	[14] &	NGC 6791	&	$-$0.08	&	0.07	&	7	&	D	&	[30] \\
NGC 4349	&	$-$0.12	&	0.04	&	5	&	D	&	[49] &	NGC 6791	&	+0.50	&	--	&	--	&	J	&	[35] \\
NGC 4349	&	$-$0.060	&	0.123	&	6	&	D	&	[61] &	NGC 6791	&	+0.05	&	0.05	&	--	&	J	&	[43] \\
NGC 4609	&	+0.05	&	0.13	&	1	&	D, W	&	[17] &	NGC 6791	&	+0.00	&	0.14	&	6	&	D	&	[49] \\
NGC 6791	&	+0.094	&	0.141	&	7	&	D	&	[61] &	Pismis 4	&	$-$0.20	&	0.08	&	2	&	D	&	[49] \\
NGC 6791	&	+0.45	&	0.04	&	--	&	C, S	&	[63] &	Pismis 4	&	$-$0.055	&	0.095	&	2	&	D	&	[61] \\
NGC 6823	&	+0.08	&	--	&	--	&	J	&	[57] &	Pismis 20	&	+0.03	&	--	&	--	&	J	&	[57] \\
NGC 6871	&	$-$0.33	&	--	&	--	&	J	&	[57] &	Ruprecht 18	&	+0.02	&	--	&	--	&	D	&	[9] \\
NGC 6913	&	$-$0.55	&	--	&	--	&	J	&	[57] &	Ruprecht 18	&	+0.08	&	--	&	1	&	D	&	[49] \\
NGC 6940	&	+0.014	&	--	&	28	&	J	&	[7] &	Ruprecht 18	&	$-$0.010	&	0.090	&	1	&	D	&	[61] \\
NGC 7044	&	+0.01	&	--	&	--	&	J	&	[57] &	Ruprecht 20	&	$-$0.29	&	0.04	&	4	&	D, W	&	[17] \\
NGC 7082	&	+0.003	&	--	&	52	&	J	&	[7] &	Ruprecht 32	&	$-$0.38	&	--	&	--	&	J	&	[57] \\
NGC 7092	&	+0.01	&	0.01	&	60	&	S	&	[44] &	Ruprecht 46	&	$-$0.22	&	--	&	--	&	D	&	[9] \\
NGC 7209	&	$-$0.01	&	0.11	&	2	&	D	&	[20] &	Ruprecht 46	&	$-$0.12	&	0.09	&	2	&	D	&	[17] \\
NGC 7209	&	$-$0.12	&	--	&	1	&	D	&	[49] &	Ruprecht 46	&	$-$0.13	&	0.18	&	2	&	D	&	[49] \\
NGC 7209	&	+0.070	&	0.050	&	2	&	D	&	[61] &	Ruprecht 79	&	$-$0.14	&	--	&	--	&	J	&	[57] \\
NGC 7209	&	$-$0.07	&	0.03	&	30	&	V	&	[64] &	Ruprecht 97	&	$-$0.59	&	--	&	1	&	D	&	[20] \\
NGC 7235	&	$-$0.62	&	--	&	--	&	J	&	[57] &	Ruprecht 97	&	$-$0.03	&	0.03	&	2	&	W	&	[25] \\
NGC 7380	&	$-$0.42	&	--	&	--	&	J	&	[57] &	Stock 2	&	$-$0.14	&	0.20	&	2	&	D	&	[20] \\
Cluster	&	[Fe/H]	&	$\sigma$[Fe/H] &	Stars	&	System	&	Ref.	&	Cluster	&	[Fe/H]	&	$\sigma$[Fe/H]	&	Stars	&	
System	&	Ref.	\\
\hline
NGC 7419	&	+0.05	&	--	&	--	&	J	&	[57] &	Tombaugh 2	&	$-$0.60	&	0.18	&	--	&	J	&	[43] \\
NGC 7789	&	$-$0.24	&	0.13	&	17	&	D	&	[49] &	Trumpler 1	&	$-$0.71	&	--	&	--	&	J	&	[57] \\
NGC 7789	&	$-$0.089	&	0.152	&	20	&	D	&	[61] &	Trumpler 10	&	$-$0.13	&	0.20	&	--	&	S	&	[37] \\
NGC 7790	&	$-$0.22	&	--	&	--	&	J	&	[57] &	Trumpler 11	&	$-$0.61	&	--	&	--	&	J	&	[57] \\
Pismis 4	&	$-$0.08	&	0.09	&	2	&	D, W	&	[17] &	Trumpler 14	&	$-$0.03	&	--	&	--	&	J	&	[57] \\
\hline
\end{tabular}
\tablefoot{Photometric systems: Caby (C) DDO (D) Johnson (J) Str{\"o}mgren (S) Vilnius (V) Washington (W). \\
Literature: [1]~Anthony-Twarog \& Twarog (\cite{Anthony-Twarog2000}); [2]~Anthony-Twarog \& Twarog (\cite{Anthony-Twarog2004}); [3]~Anthony-Twarog \& Twarog (\cite{Anthony-Twarog2006a}); 
[4]~\mbox{Anthony-Twarog} et~al. (\cite{Anthony-Twarog2005}); [5]~Anthony-Twarog et~al. (\cite{Anthony-Twarog2006b}); 
[6]~Bruntt et~al. (\cite{Bruntt1999}); [7]~Cameron (\cite{Cameron1985b}); [8]~Clari{\'a} (\cite{Claria1982}); 
[9]~Clari{\'a} (\cite{Claria1985a}); [10]~Clari{\'a} \& Lapasset (\cite{Claria1983}); [11]~Clari{\'a} \& Lapasset (\cite{Claria1985b}); [12]~Clari{\'a} \& Lapasset (\cite{Claria1986a}); 
[13]~Clari{\'a} \& Lapasset (\cite{Claria1986b}); [14]~Clari{\'a} \& Lapasset (\cite{Claria1989a});
[15]~Claria \& Mermilliod (\cite{Claria1992}); [16]~Clari{\'a} \& Minniti (\cite{Claria1988});  
[17]~Clari{\'a} et~al. (\cite{Claria1989b}); [18]~Clari{\'a} et~al. (\cite{Claria1991}); [19]~Clari{\'a} et~al. (\cite{Claria1994});
[20]~Clari{\'a} et~al. (\cite{Claria1996}); [21]~Clari{\'a} et~al. (\cite{Claria1999});  
[22]~Clari{\'a} et~al. (\cite{Claria2003}); [23]~Clari{\'a} et~al. (\cite{Claria2005}); [24]~Clari{\'a} et~al. (\cite{Claria2007}); 
[25]~Clari{\'a} et~al. (\cite{Claria2008}); [26]~Dawson (\cite{Dawson1981}); 
[27]~Eggen (\cite{Eggen1983}); [28]~Geisler \& Smith (\cite{Geisler1984}); [29]~Geisler et~al. (\cite{Geisler1992}); [30]~Janes (\cite{Janes1984a}); [31]~Janes \& Smith (\cite{Janes1984b}); 
[32]~Kaluzny \& Mazur (\cite{Kaluzny1991a}); [33]~Kaluzny \& Mazur (\cite{Kaluzny1991b});
[34]~Kaluzny \& Mazur (\cite{Kaluzny1991c}); [35]~Kaluzny \& Rucinski (\cite{Kaluzny1995}); 
[36]~Kyeong et~al. (\cite{Kyeong2008}); [37]~Lynga \& Wramdemark (\cite{Lynga1984});
[38]~McClure et~al. (\cite{McClure1981}); [39]~Mermilliod et~al. (\cite{Mermilliod2001}); 
[40]~Nissen (\cite{Nissen1980}); [41]~Nissen (\cite{Nissen1988}); [42]~Nissen et~al. (\cite{Nissen1987}); [43]~Noriega-Mendoza \& Ruelas-Mayorgo (\cite{Noriega-Mendoza1997});  
 [44]~Palous \& Hauck (\cite{Palous1986}); [45]~Parisi et~al. (\cite{Parisi2005}); [46]~Pastoriza \& Ropke (\cite{Pastoriza1983}); [47]~Paunzen et~al. (\cite{Paunzen2003}); 
[48]~Philip (\cite{Philip1976}); [49]~Piatti et~al. (\cite{Piatti1995});
[50]~Piatti et~al. (\cite{Piatti2003a}); [51]~Piatti et~al. (\cite{Piatti2003b}); [52]~Piatti et~al. (\cite{Piatti2004}); 
 [53]~Richtler (\cite{Richtler1985}); [54]~Smith (\cite{Smith1983a}); [55]~Smith \& Hesser (\cite{Smith1983b}); [56]~Sung et~al. (\cite{Sung2002}); [57]~Tadross (\cite{Tadross2003}); 
 [58]~Twarog (\cite{Twarog1983}); [59]~Twarog et~al. (\cite{Twarog1993}); [60]~Twarog et~al. (\cite{Twarog1995}); [61]~Twarog et~al. (\cite{Twarog1997}); [62]~Twarog et~al. (\cite{Twarog2006}); [63]~Twarog et~al. (\cite{Twarog2007}); [64]~Vansevicius et~al. (\cite{Vansevicius1997}).}
\end{table}